\begin{table}%t1
%\centering
\par
\caption{\label{tab:entries1}SPECFIND V1.0 catalogue entries.}
\begin{tabular}{lcrrrrcl}
\hline \hline \noalign{\smallskip}
Catalog & I/S\tablefootmark{1} & Frequency & Resolution & $S_{\rm min}$ & Number of & Percentage\tablefootmark{2} &  Reference \\
name &  & (MHz) & (arcmin) & (mJy) & sources &  & \\
\hline\noalign{\smallskip}
JVAS & I &  8400 &  $5.5 \times 10^{-3}$   & 30 & 2246 & 72 &  Patnaik et~al. (\cite{Patnaik1992})  \\
 & & & & & & & Browne et~al. (\cite{Browne1998}) \\
 & & & & & & & Wilkinson et~al. (\cite{Wilkinson1998}) \\
GB6  & S &  4850 &  3.5      &  18 & 75162  & 59 & Gregory et~al. (\cite{Gregory1996}) \\
87GB & S &  4850 &  3.5      &  25 & 54~579 & 64 & Gregory \& Condon (\cite{Gregory1991}) \\
BWE  & S &  4850 &  3.5      &  25 & 53~522  & 61 & Becker et~al. (\cite{Becker1991}) \\
PMN  & S &  4850 &  3.5      &  20 & 50~814  & 30 & Wright et~al. (\cite{Wright1994}, \cite{Wright1996}) \\
 & & & & & & & Griffith et~al. (\cite{Griffith1994}, \cite{Griffith1995}) \\
MITG & S &  4850 &  2.8      &  40 & 24~180 & 52 & Bennett et~al. (\cite{Bennett1986}) \\
 & & & & & & & Langston et~al. (\cite{Langston1990}) \\
 & & & & & & & Griffith et~al. (\cite{Griffith1990}, \cite{Griffith1991}) \\
PKS  & S &  2700 &  8.0      &  50 & 8264  & 65 & Otrupcek \& Wright (\cite{Otrupcek1991}) \\
F3R & S &  2700 &  4.3      &  40 & 6495 & 61 & F\"{u}rst et~al. (\cite{Furst1990})  \\
FIRST & I & 1400 &  0.083  &  1 & 811~117 & 1.5 & White et~al. (\cite{White1998}) \\
NVSS  & I & 1400 &  0.75     &  2 & 1773484 & 3.6 & Condon et~al. (\cite{Condon1998}) \\
WB   &  S & 1400 & 10.       & 100 &  31524  & 60  & White \& Becker (\cite{White1992}) \\
SUMSS & I &  843  &  0.75     &  8 & 134~870 & 1.8 & Mauch et~al. (\cite{Mauch2003})  \\
B2  & I &  408  &  8.0       & 250 &  9929 & 72 & Colla et~al. (\cite{Colla1970}, \cite{Colla1972}, \cite{Colla1973})  \\
 & & & & & & & Fanti et~al. (\cite{Fanti1974}) \\
B3 & I & 408  &  5.0      &  100 & 13~340 & 66 & Ficarra et~al. (\cite{Ficarra1985}) \\
MRC & I & 408 &  3.0      &  700 & 12~141 & 73 &  Douglas et~al. (\cite{Douglas1996})   \\
TXS  & I &  365  &  0.1      &  250 & 66~841 & 57 & Large et~al. (\cite{Large1991}) \\
WISH & I & 325  &  0.9      &  10 & 90~357 & 8.4 & de Breuck et~al. (\cite{Breuck2002}) \\
WENSS & I & 325  &  0.9      &  18 & 229~420 & 17 & Rengelink et~al. (\cite{Rengelink1997}) \\
MIYUN & I & 232  &  3.8      &  100 & 34~426 & 40 & Zhang et~al. (\cite{Zhang1997}) \\
4C  & I &  178  &  11.5     &  2000 & 4844 & 53 & Pilkington \& Scott (\cite{Pilkington1965})  \\
 & & & & & & & Gower et~al. (\cite{Gower1967}) \\
3CR & I & 178  &  6.0       &  5000 & 327 & 31 & Bennet (\cite{Bennett1962}) \\
3C   & I & 159  &  10.0      &  7000 & 470 & 3.4 & Edge et~al. (\cite{Edge1959}) \\ 
\hline
\end{tabular}
\tablefoot{\tablefoottext{1}{S: single dish; I: interferometer.}
\tablefoottext{2}{Percentage of sources with identified spectrum.}}
\end{table}